Author's response to reviews

Title: Adult systemic cat scratch disease associated with therapy for hepatitis C

Authors:

Zahida Bhatti (zbhatti@hotmail.com)
Charles S Berenson (berenson@acsu.buffalo.edu)

Version: 5 Date: 22 February 2007

Author's response to reviews: see over
22 February 2007
BioMed Central Editorial Production Team
BMC-series Journals
BioMed Central

Dear BioMed Central Editorial Production Team:

We are delighted with the acceptance of our manuscript, “Adult systemic cat scratch disease following therapy for hepatitis C” by Z. Bhatti and C. S. Berenson (MS: 1554525631115588) to BMC Infectious Diseases. The newly submitted version of the manuscript conforms to each of the formatting changes that we were asked us to make. Please let me know if there is anything further that you require.

Sincerely yours,

Charles S. Berenson